

Article



Unsupervised Learning for Product Use Activity Recognition: An Exploratory Study of a "Chatty Device"

Mike Lakoju^{1,*} ^(D), Nemitari Ajienka², M. Ahmadieh Khanesar³ ^(D), Pete Burnap⁴ and David T. Branson³

- ¹ Cardiff School of Technologies, Cardiff Metropolitan University, Western Avenue, Cardiff CF5 2YB, UK
- ² Department of Computer Science, Nottingham Trent University, Nottingham NG11 8NS, UK; nemitari.ajienka@ntu.ac.uk
- ³ Faculty of Engineering, The University of Nottingham, University Park, Nottingham NG7 2RD, UK; ezzma5@exmail.nottingham.ac.uk (M.A.K.); ezzdtb@exmail.nottingham.ac.uk (D.T.B.)
- School of Computer Science & Informatics, Cardiff University, Wales CF10 3AT, UK; burnapp@cardiff.ac.uk
- Correspondence: mLakoju@cardiffmet.ac.uk

Abstract: To create products that are better fit for purpose, manufacturers require new methods for gaining insights into product experience in the wild at scale. "Chatty Factories" is a concept that explores the transformative potential of placing IoT-enabled data-driven systems at the core of design and manufacturing processes, aligned to the Industry 4.0 paradigm. In this paper, we propose a model that enables new forms of agile engineering product development via "chatty" products. Products relay their "experiences" from the consumer world back to designers and product engineers through the mediation provided by embedded sensors, IoT, and data-driven design tools. Our model aims to identify product "experiences" to support the insights into product use. To this end, we create an experiment to: (i) collect sensor data at 100 Hz sampling rate from a "Chatty device" (device with sensors) for six common everyday activities that drive produce experience: standing, walking, sitting, dropping and picking up of the device, placing the device stationary on a side table, and a vibrating surface; (ii) pre-process and manually label the product use activity data; (iii) compare a total of four Unsupervised Machine Learning models (three classic and the fuzzy C-means algorithm) for product use activity recognition for each unique sensor; and (iv) present and discuss our findings. The empirical results demonstrate the feasibility of applying unsupervised machine learning algorithms for clustering product use activity. The highest obtained F-measure is 0.87, and MCC of 0.84, when the Fuzzy C-means algorithm is applied for clustering, outperforming the other three algorithms applied.

Keywords: Chatty Factories; Industry 4.0; machine learning; IoT; product use activity; sensors; clustering

1. Introduction

As "Industry 4.0" draws significant attention, a large number of organizations are seeking novel methods and techniques of acquiring and processing huge amounts of sensor data. This is considered the birth of a new industrial revolution [1,2]. The first industrial Revolution came as a result of the mechanization of production, which affected product volume dimension. The second Revolution altered the industry by leveraging on the advent of electricity and mass production. The third Revolution was characterized by the adoption of Information Technology (IT) and electronics to process automation [3,4]. More recent discussions are suggestive of the birth of the fourth Industrial Revolution, with the emergence of "Industry 4.0" and "Smart Factory". By implication, product design and manufacturing within the Industry 4.0 paradigm is now more data-driven.

This most recent revolution in technology offers enormous benefits to product design teams [5,6], such as the capability to leverage sensors to acquire data directly from products. Understanding customers' needs, satisfaction, experience, etc., form a critical aspect of



Citation: Lakoju, M.; Ajienka, N.; Khanesar, M.A.; Burnap, P.; Branson, D.T. Unsupervised Learning for Product Use Activity Recognition: An Exploratory Study of a "Chatty Device". *Sensors* **2021**, *21*, 4991. https://doi.org/10.3390/s21154991

Academic Editor: Raffaele Gravina

Received: 24 May 2021 Accepted: 19 July 2021 Published: 22 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). creating products that satisfy segments of customers and increasing profitability. Manual processes used to collect customer's feedback can now be automated by leveraging sensors. This, in turn, enables the creation and automation of data-driven processes/strategies for acquiring meaningful product use information that could influence the creation of new products or the enhancements of existing ones.

Advancements in sensor technology have exponentially increased the number of Internet-connected products being produced in factories all around the world today. This, in turn, creates the possibility of collecting massive amounts of product use data that can now be leveraged by product design teams [5–7]. This is akin to Opresnik et al.'s [8] opinion that gaining an understanding of product use activities is a data-driven process of information, which is a feedback loop of collecting, storing, and then analyzing data from the product-users with a primary goal to discover and identify usage patterns that inform design suggestions. Therefore, design teams are equipped to tailor products to customers' needs and wants.

Exploring product-use data strategically gives design engineers a competitive advantage by enabling them to uncover patterns, innovative insights, and knowledge through data-driven design [9]. Conventional product design involves a technical decision process that is characterised by the intrinsic performance of the product [10]. Typically, hypothetical market studies are acquired as a collection of design specifications or based on experiential knowledge, which is subjective and consequently deviates from true customer satisfaction [11]. Interviews, user experiments, focus groups, etc., were a few of the typical methods used in acquiring customer requirements. However, this was limiting because only the customers' actions and extrinsic behaviors could be captured. Ultimately, this failed to provide latent customer needs and intrinsic requirements for product performances [12].

The data-driven analysis approach towards product design is grounded on the premise that, when manufacturers know how customers are using the products, they can tailor their products much better to actual needs. This enables design decisions to be based on facts and not assumptions [9]

The Chatty Factories concept focuses on the opportunity to collect data from IoTenabled sensors embedded in products (Chatty devices) during real-time use by consumers, explores how that data might be immediately transferred into usable information to inform design, and considers what characteristics of the manufacturing environment might optimise the response to such data. Figure 1 illustrates the Chatty Factories concept. A detailed background can be found in Reference [13]. In summary, the concept offers the potential to accelerate product refinement through a radical product interruption of 'consumer sovereignty' based around surveys and market research, to 'use sovereignty' and an embedded understanding of consumer behavior—making products that are fit for purpose based on how they are used. Achieving continuous product refinement in response to real-time data on product use requires digital tools that can interact with physical and virtual life. The "Data Annotation" block in Figure 1 is one of the key phases of the proposed data driven model which requires the use of machine learning models to identify anomalies, patterns, and points of interest in the data. This is the area we address in this paper.

Sensor data streaming from the wild can hold key information about how the product is used by consumers across different locations and cultures. Gathering data from use in the wild can limit the need for current product research studies, such as post-hoc lab studies and user experience surveys. Furthermore, in-situ research methods can capture the intricate and messy relationships between people and products, which may be difficult to assess through lab-studies [14].

Due to the volume of real-time data created by the sensors and the fact that products will be used by a large scale of customers, it becomes necessary to employ the use of unsupervised machine learning models to identify anomalies, patterns, and points of interests at scale. In general, labeling real time activity data is very difficult. The main reason is that, within a single experiment, there may be various types of activities which makes labeling data very difficult, if not impossible. To this end, this paper compares a number of unsupervised machine learning models. Their successful implementation provides evidence to use these approaches for future unlabeled data used for dealing with product activity analysis in a Chatty Factories framework.



Figure 1. The Chatty Factories vision.

Today, the competitive business environment is putting more pressure on manufacturers to constantly improve/manufacture products that satisfy customer demands, usually through changes in design and functionality, relying primarily on continuous analysis of large amounts of data. Product data can be obtained from varying sources; however, technology advancements in sensor technology has inspired many intelligent products— "product embedded information devices (PEID)" [15]. Furthermore, The Internet of Things (IoT), being an ecosystem of such inter-connected devices, makes data transfer even more possible, thus enabling more efficient product lifecycle data. Products retrofitted with a wide array of sensors improve the product's capability to store, compute, and communicate data (chatty devices), for example, smartphones, tablets, smart wearables, water monitor, etc. [16–18].

Leveraging product use data/PEID within a well thought out data-driven methodology becomes a more precise customer feedback method [15] than the manual processes of collecting customer feedback previously utilised in product design [19]. It, therefore, becomes imperative to build on work done around activity recognition that sought to understand people's daily lives. In this paper, we seek to understand actual product experiences, using activity leveraging unsupervised learning methods.

Activity recognition is a well explored research topic. Researchers have typically focused on Human activity recognition (HAR) because it provides insight into understanding the daily lives of people inferred from analysis carried out from raw sensor inputs [20]. Scholars have successfully investigated Human Activity Recognition in areas, such as home behavior analysis [21], gesture recognition [22], gait analysis [23], video surveillance [24], etc.

Real-time feedback acquired from the actual product-use through sensors embedded in the products would be extremely valuable for product designers to understand how the product is being used. Human behavior identification/classification by interaction with the product can play a crucial part in understanding how the product is being used. But so can producing experiences beyond human activity. For example, a product may roll off an uneven surface, or be transported in a rough manner—neither are always under the direct control of the human. We argue that this additional information about how products "experience the world" will potentially offer new insights for product design teams.

This study focuses on product activity for the purposes of understanding experiences in the wild, and informing future versions of its design and manufacture. It seeks to identify actual product use activity by creating experiments using a "Chatty Device" with inbuilt sensors, such as Accelerometer, Orientation sensor, Magnetic field, and Angular Velocity. Since data used in this study suffers from high dimension, a dimension reduction stage is required. Hence, unsupervised learning approaches because of their power to reduce the data dimension are the preferred approach in this paper [25]. A sampling rate of 100 Hz was used to collect data around common product experiences, such as vibrations, being stationary, drops, and pickups—alongside human activities, while using products, including walking, standing, and sitting. A total of about 148,778 data points were collected.

To our knowledge, general research in activity classification has focused on areas, such as human activity classification, health-related studies, smart homes, etc., while we explore and investigate product use activity classification in the wild for the first time. We approach the problem as an unsupervised learning problem, even though we have created label data. Our rationale is to provide verifiable evidence that allows us to map the actual product use activity to the identified cluster detected by the algorithm.

In this study, we utilize securely and ethically collected product-use big data to investigate how unsupervised learning algorithms can be used to understand at scale, product use in the wild. This is key to the Chatty Factories concept, which is an innovative industry 4.0 model. In this framework, human behavior identification is crucial as it will provide valuable information about how and where the products is used for further analysis by the design team. Product behavior associated with different human activity in terms of vibrations and orientations met by the product provide valuable insight for the design team for possible reinforcement of the product to deal with the situations met by the product as a result of different product use activities. We log sensor data, pre-process it and label the data according to the product use activity carried out. The activity labels created are only for validating the clusters mapped to a specific activity. This helps to evaluate the effectiveness of the clustering method used, thus enabling the authors to justify the approach of using Unsupervised Machine learning to identify unknown product use activities. The standpoint of the authors is that Unsupervised Machine Learning methods can be used effectively to cluster similar product use activities, even if initially unknown. An unknown product use activity "X" can be discovered, and, after an ethnographic intervention, a label can be subsequently assigned to "X", for example, "A drop".

The following are the main contributions of this study:

- the first approach to detect discrete product use activities in the wild (vibrations, being stationary, drops, and pickups), enhancing previous approaches that focus only on human activity (walking, standing, and sitting). Product behavior within each product use activity can then be studied by the design team for further investigation and possible modifications for the product;
- the use of the Fuzzy C-means algorithm to effectively and efficiently detect product use activities. The behavior of product associated with product activities would then be easy to study as, instead of dealing with large amount of data within each cluster, the center of cluster can be studied; and
- a novel (publicly available) curated and manually verified dataset consisting of actual product use activity allowing researchers to conduct further studies in this domain and compare machine learning results (Dataset is available at: https://doi.org/10.6 084/m9.figshare.11475252.v1, accessed on 3 December 2020) curated and manually verified dataset consisting of actual product use activity (for further comparative studies) allowing researchers to conduct further studies in this domain and compare machine learning results).

The rest of the paper is structured as follows: Section 2 provides an overview of related studies on activity detection. Section 3 discusses the materials used in the study, such as the sensors, dataset, and product use activities. Section 4 is the Methodology section, which describes the empirical approach used to carry out the product use activity clustering using unsupervised machine learning methods, including the data pre-processing and independent feature selection process. Furthermore, Section 4.4 summarizes the results and discusses the findings and provides further empirical insights. Section 5 describes the unsupervised machine learning results, while Section 6 concludes the paper.

2. Related Work

This section provides an overview of related research on the application of machine learning techniques for activity detection.

Dealing with product-use activity classification/identification at scale requires the application of machine learning to identify those patterns and anomalies. There are many and varied studies on "human activity classification"; however, the context of primarily focusing on "product-use activity classification" is limited. This has prompted a key objective of this study. It is believed that detecting activities, like picking up the device, dropping the device, or exposing the device to a vibrating environment, etc., can provide useful product design information [8]. For instance, Ankita et al. [26] focused primarily on improving human activity recognition by combining convolutional layers, long shortterm memory (LSTM), and the deep learning neural network to extract key features in an automated way and then categorize them with some model attributes. Essentially, their study did not focus on product activity recognition; they tested their model on a publicly available dataset of UCI-HAR for Samsung Galaxy S2, which captures various human activities, such as walking, walking upstairs and downstairs, sitting, standing, and laying. They did not look at activities that could impact a product, like vibrations within the environment. Russell et al. [27] also primarily focused on the context of human activity recognition. They set out to create and validate a field-based data collection and assessment method to aid human activity recognition in the mountains with terrain and fatigue variations. An output from their study was an unsupervised labeled dataset of various longterm field-based activities, such as obstacle climb, stand, run, walk, and lay. Furthermore, they were keen on the human context rather than product classification. As such, they were evaluating Fatigue levels by modulating between rested to physical exhaustion.

In this section, we review literature, focusing on activities, such as "vibration", "drop", and "pickup", and others, like "standing", "walking", "sitting", and "being stationary", in order to determine whether existing research on activity detection has achieved the detection of these activities particularly within the context of application for product-activity classification. Table 1 summarizes some of the gaps in existing literature.

While still focusing on human activity recognition, Jun and Choi [28], focused on a slightly different area. They argue that lots of studies have been done recognizing Adult human activities; however, HAR would be beneficial for the safety and wellness of newborns or infants because they have poor verbal communication. To this end, they focus their research on newborn and infant activity recognition by analyzing accelerometer data by attaching sensors to the body. They classify four types of activities: sleeping, moving in normal condition, moving in agony, and movement by an external force. Furthermore, they took an unsupervised learning approach torwards recognizing activies leveraging an end-to-end deep model using autoencoder and k-means clustering. Similarly, their study did not take into consideration product activity recognition.

Gao et al. [29] carried out a study on human activity recognition based on Stacking Demonising Autoencoder (SDAE) and LightGBM (LGB). Their approach uses the SDAE to remove noise from the raw sensor data to extract valid characteristics of the data with unsupervised learning. At the same time, the LGB was applied to distinguish the inherent feature dependencies among categories for accurate human activity recognition. They test their algorithm on activities, such as walking, going up and down an Elevator and Escalator, standing, sitting, lying, etc. The study did not consider product-use recognition nor evaluate activities, like vibrations from the environment, that could affect a product.

Wyatt, Philipose, and Choudhury [30] considered unsupervised learning methods in their study positing activities can be differentiated or identified by the objects used. Presented with the names of these objects, they created models of activities from the web, and proposed that the models can distinguish twenty-six (26) activities. However, they did not consider activities, like vibrations, drop and picking up, or stationary position. Huynh [31] proposed an unsupervised learning model that deals with multiple time scales, by leveraging multiple eigenspaces. In their study, activities, such as walking, standing, and sitting, were recognized with four inertial sensors. Li and Dustdar [32] conducted experiments with the anticipation of adopting unsupervised learning methods in activity recognition. They put forward a hybrid process that fuses activity classification with subspace clustering in a bid to handle the curse of dimensionality, but their research did not look into specific activities, like drop and pickup, vibrations, etc., and Vandewiele and Motamed [33] looked into a smart home environment with a network of sensors. They suggest that an unsupervised learning approach would be useful for activity recognition in such an environment; however, they did not consider activities, such as drop and pickup, vibrations, walking, stand-in, sitting, or stationary surface. They also did not focus on the context of application to product-use activity identification.

Trabelsi [34] put forward an unsupervised learning method for human activity recognition by using three accelerometers which were attached to the right thigh, left ankle, and chest. They used the Hidden Markov Model Regression and proposed that their model substantially outperformed other unsupervised learning methods and was closely comparable to supervised learning methods. However, for the application of the algorithms, it is assumed that the number of activities is known. They also did not consider activities, like vibrations or drop and picking up. Furthermore, a number of scholars have focused on human activity recognition by leveraging a combination of sensors which are embedded in cameras [35], mobile devices, wearable computers, etc. Accelerometers are one of the most frequently used sensors for activity recognition because its functionality helps in measuring activities performed by users [36]. Some studies employ the use of multiple accelerometers attached to different locations of the human body. For example, Reference [37] ran experiments that used two or three uniaxial accelerometers to differentiate various activities, which include people lying down, standing, sitting, descending and ascending stairs, and also cycling. They, however, fail to consider activities, like vibration or drop and pickup. Arminian [38] investigated the possibility of using two accelerometers, one attached to the rear of the thigh and the other on the chest. Foerster and Fahrenberg [39] placed three uniaxial accelerometers on the sternum, and two uniaxial accelerometers on the right and left thighs to identify four basic activities (moving, standing, sitting, and lying down).

According to Reference [40], human activity detection can be done using static platforms or wearable platforms. Two major static platforms to detect human motions are marker-based motion capture system [41] and force plate system [42]. The combination of these two static platforms may also be used for human activity detection purposes. Other than static platforms, wearable platforms can be used to detect human motion. Examples of wearable platforms are accelerometers [43], gyroscopes, magnetometers, electro/flexible goniometers [44], and sensing fabrics [45]. The combination of these sensors may also be used for activity detection purposes [40]. In this study, the combination of accelerometers, gyroscopes, and magnetometers is used for product activity detection. The fact that all of these sensors are available under iPhone makes it a cheap product activity detection method.

Previous research typically employed machine learning methods to investigate activity recognition; however, Table 1 highlights some gaps based on some specific activities that could potentially facilitate product design. This study focuses on activity recognition of product use. It presents the application of unsupervised machine learning models on a novel real-time product use activity dataset derived from sensors embedded in a "chatty"

Paper	Activities						Application in Product-Use Activity Identification
	Drop & Pickup	Vibrations	Walking	Standing	Sitting	Stationary Position	
[23,26,29,31,34,37-39,46-72],	0	0	Х	Х	Х	0	0
[27,73–78]	0	0	Х	Х	0	0	0
[23,79–86]	0	0	Х	0	0	0	0
[33,87–89]	0	0	0	0	0	0	0
[90]	0	0	0	0	Х	0	0
[35,91]	0	0	Х	0	Х	0	0
[30,92]	0	0	0	Х	0	0	0
This Paper	Х	Х	Х	Х	Х	Х	Х

product. This will underpin an understanding of how the product is used at scale and could inform product design modification in a way hitherto not possible.

Table 1. Summary of activity recognition on vibration, drop and pickup, standing, sitting, walking, and stationary position.

3. Materials

For this study, a dataset was created by conducting expected product use activities, such as placing the device on a vibrating surface, dropping and picking up of the device, walking with the device, sitting with the device, placing the device stationary on a side table, and standing with the device. A 6th generation Apple iPad was used as the "Chatty device" for this study because it has all the inbuilt sensors that enable the research team to measure and detect these activities. Modern mobile iOS and Android devices are equipped with a rich variety of sensors, thus presenting an ideal medium to collect data for research. Apple devices typically offer a number of functionalities that leverage the inbuilt sensors in the devices. These include proximity sensors, accelerometers, ambient light sensors, gyroscope, and more [93], which, for example, can enable a user shake the device to undo an action or dim the screen when the device is held at face level, etc. Figure 2 illustrates the flow of the experiment.



Figure 2. Flow of work.

The data was collected from the following four in-built sensors.

- Acceleration: 3-axes acceleration data,
- Orientation: Azimuth, Pitch, and Roll,
- Angular Velocity: 3-axes gyroscopes for rotational motion,
- Magnetic Field: 3-axes magnetic field.

Acceleration on 3-axes: This sensor is used to measure the linear acceleration of the device. The sensor has the capability to function in two modes ± 2 g and ± 8 g. Both modes can sample data at 400 Mhz or at 100 Mhz. Apple uses a default of ± 2 g mode (100 Mhz) [94].

Azimuth, roll, pitch (Orientation): Another sensor pre-built into the device is a vibrational gyroscope. It functions by utilizing a plate called a "proof mass". When a signal is applied to capacitor plates, it oscillates. When an individual displaces the device in a rotational manner, the proof mass is displaced in the X, Y, Z directions an ASIC processor measures the capacitance change of the plates [94].

Magnetic Field on 3-axes: This sensor can measure the strength of the magnetic field surrounding the device. Hence, the device is equipped with the ability to determine its "heading" in line with the geomagnetic North Pole, thus acting as a digital compass [94].

Angular Velocity on 3-axes: The angular velocity also known as rotational velocity, can be explained as the rate of velocity that an object or a particle rotates around a defined point at a given time.

4. Methodology

In this section, we describe the methodology adopted for conducting the experiments/study.

4.1. Data Acquisition

The MATLAB Mobile application was used to stream data from the in-built sensors of the iPad (Chatty-device). MATLAB is a purposed built scientific environment for running experiments. It is created for engineers and scientists with a robust set of Apps and toolboxes prepackaged and rigorously tested for a broad range of scientific and engineering applications [95]. The MATLAB mobile app application is convenient and freely available for individuals with an academic email. Hence, the research team decided to use the MATLAB mobile app, which runs on both Android and iOS devices and is user friendly. Some open-source applications can also collect data from mobile devices; however, the majority of them are platform-dependent and, most times, not free. The product use data was streamed into a MathWorks Cloud drive account at a sampling rate of 100 Hz per activity. The research team collected about four (4) minutes of sensor data for every activity through a total of a total of 148,778 data points. The following steps were taken:

- 1. The MATLAB mobile app was installed and configured on the Chatty device through the Apple App store.
- 2. A registered user account was used, and the device was connected to MathWorks Cloud account.
- 3. Using the MATLAB Mobile interface, the sensors were turned "on".
- 4. A sampling rate of 100Hz was selected, and the log button was used to initiate recording of the sensor data readings.
- 5. After the expected duration, the logged results were saved to the Cloud and given file names which corresponded to the activities carried out.

Two subjects (an adult female and male) held the device while carrying out the product use activities—dropping and picking up, standing, walking, and sitting. Product use data was also collected while the device was placed on a vibrating and non-vibrating stationary surface. Figure 3 highlights the acceleration plots for the three axes (X, Y, Z) for each of the six activities. Walking is a more energetic activity so when compared with the others, its X, Y, Z values vary significantly. Standing, sitting, device on vibrating surface, and stationary on a side table are more dormant activities; thus, their X, Y, and Z values are almost constant. The drop and pickup activity plots indicate sudden spikes in the signals which come as a result of the sudden drop events that occurred. One interesting thing to observe is that, for walking, standing, sitting, vibrating surface, and stationary on a side table, the "Z" values have the larges accelerations. This is because the force of gravity affects the entire acceleration in the direction of the center of Earth. Figure 4 highlights the angular velocity for the three axes (X, Y, and Z) for each of the six activities. Large spikes can also be observed on the drop and pickup activity plot, which indicates the sudden drop events that occurred. For walking and drop and pickup activities, the X, Y, and Z values significantly change. For standing, sitting, stationary on a side table and vibrating surface, the X, Y, and Z values are much smaller (<0.3 rad/s).



Figure 3. Acceleration signals for the six activities.



Figure 4. Angular velocity signals for the six activities.

4.2. Data Pre-Processing and Feature Selection

All the logged product use sensor data was stored in the MathWorks cloud account with a ".mat" file extension. The files were downloaded and imported into MATLAB desktop application. MATLAB allows for variables to be created and used to house data. Hence, a variable was created for each recorded sensor activity. The variable contained the time-stamp and X, Y, Z axis data for each of the sensors. A MATLAB timetable file for each of the sensors was exported to a CSV file and labeled according to each product use activity carried out.

To create the Labels for each activity and assign them to their respective data points, the research team had to preprocess each data table independently before combining them. The MATLAB sensor log file that was downloaded from cloud drive for each activity held the sensor reading for the Accelerometer, Magnetic-field, Orientation, and Angular Velocity sensors. Each sensor Table had timestamp, X, Y, and Z data readings. All sensor tables where then concatenated by columns and then a new column with header "Activity" was created and assigned a value corresponding to the name of the activity carried out. This was done iteratively for each activity distinctively with their labels assigned accordingly. Finally, the research team merged all tables into one master file.

A master data CSV file was created using Microsoft Excel. It contains the combined set of sensor readings of the experiments which were arranged according to the sensors following the sequence of the activities carried out. The timestamp variables were treated as data-time variables on excel and stored accordingly. The CSV file was imported into WEKA for pre-processing and feature selection. WEKA is an open-source data analysis software [96,97] (The tool is available at: http://www.cs.waikato.ac.nz/ml/weka/index. html, accessed on 10 December 2020.) developed by the University of Waikato and has been used with its default setting [96–100].

A total of sixteen (16) independent variables were collected including the following: Acceleration X, Acceleration Y, Acceleration Z, Timestamp-Acceleration, Angular Velocity X, Angular Velocity Y, Angular Velocity Z, Timestamp-Angular Velocity, Magnetic Field X, Magnetic Field Y, Magnetic Field Z, Timestamp-Magnetic Field, Orientation X, Orientation Y, Orientation Z, and Timestamp-Orientation.

Each sensor is unique and, thus, may have a slightly different sampling rate. To this end, we collected data for each of the four sensors and also recorded their respective timestamps to each data-point.

As part of the data preparation exercise, an activity column was created to capture each specific activity. It contained the labeled activities carried out in the experiment, which also represents the dependent or response variable for the experiment. These are the class variables (product use activities) that we seek to detect with the unsupervised learning algorithms

The response variables or classes are six-fold and are as follows:

- dropping and picking up,
- vibrating,
- non-vibrating stationary surface,
- standing,
- walking, and
- sitting.

In prior research [101], the CfsSubsetEval feature evaluator in WEKA has been identified as the evaluator with the best performance for feature selection using its search methods (BestFirst, Ranker search, and Forward selection). In order to remove noisy features from the experiment or features with a weak predictive power, the *BestFirst* feature selection method of the CfsSubsetEval evaluator in WEKA was used to identify a subset of features that have a high predictive ability in relation to the dependent or response variable (product use Activity). In other words, it identifies a subset of features with a high correlation with the classes but low intercorrelation. This feature selection algorithm searches the space of attribute subsets by greedy hill climbing augmented with a backtracking facility. In addition to the BestFirst technique, a 10-fold cross validation technique was used to identify the features that maintain a high predictive power in all 10-folds. This technique involves randomly dividing the dataset into k groups or folds of approximately equal size. The first fold is kept for testing and the model is trained on k - 1 folds.

As a result of the feature selection process, the eight (8) features or independent attributes with the highest predictive power and used in the experiment are as follows: Acceleration Y, Angular Velocity X, Angular Velocity Y, Angular Velocity Z, Magnetic Field X, Orientation X, Orientation Y, and Orientation Z.

4.3. Finding an Optimal Number of Clusters

This study is aimed at identifying suitable machine learning models for the identification of product use activity in order to relay product usage insights back to product designers, engineers, and manufacturers. This will aid the enhancement of products used in the wild, as well as user satisfaction.

While it is known in advance what activities are in the collected dataset for this study (labeled dataset which should naturally constitute a classification or supervised machine learning problem) having collected the dataset in a controlled environment (as described in Section 3), products are generally used in the wild and the distinct usage activities will not be predetermined. Based on this premise, we want to identify suitable (unsupervised) clustering techniques that can identify product use activities sharing similar attributes and evaluate the performance of these clustering techniques using external performance functions based on the labels (classes) in our dataset. The labels in our dataset are, therefore, solely used to evaluate the accuracy of the clustering algorithms applied in this study, to provide an estimate of how well they will perform on product use datasets gathered from the wild. As we are resolving a clustering problem, it is imperative to identify an optimal number of clusters to use in our experiments, given that the number of product use activities will be unknown in the wild.

Among different methods to find the optimal number of clusters, the elbow method is selected [102]. In this algorithm, the center of clusters is estimated using the K-means clustering algorithm (to perform the clustering) and then the *Within-Cluster-Sum-of-Squares* (WCSS) is estimated for each cluster (WCSS is the sum of squares of the distances of each data point in all clusters to their respective centroids.) [103,104].

The fundamental idea of the elbow method is to adopt the square of the distance between the sample points in each cluster and the centroid of the cluster to give a series of K values. The Within-Cluster-Sum-of-Squares (WCSS) is adopted as a performance indicator and we can iterate over the K-value and calculate the WCSS [105]. Smaller values indicate that each cluster is more convergent. When the number of clusters is set to approach the number of real clusters, WCSS shows a rapid decline. When the number of clusters exceeds the number of real clusters, WCSS will continue to decline, but it will quickly become slower. The K value can be better determined by plotting the K-WCSS curve and by finding the inflection point where the curve yields an angle.

The elbow method is adopted as it is visual, more intuitive, and examines each data point within each cluster in comparison to the cluster's centroid. In addition, the elbow method is mainly a decision rule, compared to other methods, such as silhouette, which is a metric used for validation while clustering. As such, it can be used in combination with the elbow method. Based on this premise, the elbow method and the silhouette method are not alternatives to each other for finding the optimal K. Rather, they are tools which can complement each other for a more confident decision [102].

The corresponding graph is plotted with respect to the number of clusters and WCSS (as in Figure 5). The optimal number of clusters is the point at which increasing the number of clusters does not improve the clustering index considerably. Figure 5 demonstrates corresponding graph for all six classes or product usage activities in our dataset. Considering this figure, the optimal number of clusters selected is 6 for all classes.



Figure 5. Elbow method to select number of clusters.

4.4. Experiments

The unsupervised machine learning experiments were carried out based on the knowledge of the optimal number of activities (k) in order to gain a good understanding of the performance of unsupervised machine learning methods on each of the sensors for product use activity recognition. Given that the optimal k was known (as described in Section 4.3, k = 6), which is also equivalent to the number of classes in this case (the activity response variable), we used the "Classes to clusters evaluation" clustering mode in WEKA and applied the following three classic unsupervised machine learning algorithms (with k = 6, using the default options for each clustering algorithm [106]): K-means, Expectation-Maximization (EM), and Farthest First. These three unsupervised machine learning algorithms have also been explored in a prior study.

In addition, we carried out another experiment using fuzzy C-means algorithm [107,108], which was implemented in Python using skfuzzy package as they are unavailable in WEKA. We selected this additional algorithm because it enables us to consider activity as belonging to more than one discrete cluster. This means multiple concurrent activities (e.g., walking and dropping) can belong to different clusters to different degrees, given the activity is performed in real time. We anticipate that defining cluster memberships to different degrees will improve the performance of an activity-focused clustering approach. In addition, we wanted to conduct a more robust and comprehensive accuracy comparison of unsupervised machine learning algorithms with different features, complexities, and strengths. All four algorithms are further described below.

4.4.1. WEKA

In WEKA, we used the Classes to clusters evaluation clustering mode when performing the clustering. Using the Classes to clusters evaluation mode, in the training phase, WEKA ignores the classes or label attribute and generates the clustering based on a selected clustering algorithm. The testing phase then proceeds after that. During the testing phase, classes are assigned to the clusters, based on the majority value of the class attribute within each cluster. Finally, the classification error is computed. Based on this assignment a corresponding confusion matrix is derived, from which we can compute the evaluation metrics (precision, recall and f-measure) described in Section 4.5.

K-Means

This algorithm is a well-used clustering algorithm [109]. It is a distance-based technique which functions in an iterative manner. At the initiation of the *K*-means algorithm, *k* objects would be randomly selected to be the centers of the clusters. All objects are then grouped into k clusters based on the minimum squared-error criterion (Only one assignment to one center is possible. If multiple centers have the same distance to the object, a random one would be chosen.). This helps to measure the distance between an object and the cluster center. At this point, the new mean is calculated for each cluster and then a new iteration process is carried out until the cluster centers remain the same [110,111].

Let $X = \{x_i\}$, i = 1, 2, ..., n be the n objects to be clustered, $S = S_1, S_2, ..., S_k$ is the set of clusters. Let μ be the mean of cluster S_i . The squared-error between μ_i and the objects in cluster S_i is defined as $WCSS(S_i) = \sum_{X_j \in S_i} ||X_j - \mu_i||^2$. The K-means algorithm aims to minimize the sum of the squared error over all k clusters, that is $min(WCSS(S)) = argmin_s \sum_{i=1}^k \sum_{X_j \in S_i} ||X_j - \mu_i||^2$, where WCSS denotes the sum of the squared error in the inner-cluster. A drawback to K-means algorithm is that it requires the specification of the number of "k" (clusters) beforehand and also and the inappropriateness for discovering non-convex clusters [112].

Expectation-Maximization (EM)

This algorithm functions by determining the members of each data object according to a probability [113]. EM consists of maximization steps and expectation. The cluster probability of each object is calculated by the expectation step while the maximization step calculates the distribution parameters that maximize the likelihood of the distributions given the data [110,111]. The EM algorithm has one notable advantage, which is being stable and easy to implement. However, a drawback is that it is slow to converge and intractable expectation and maximization steps [114]. This is represented by the formulas:

$$Q(\theta \mid \theta^{(t)}) E_Z \mid_{X,\theta} {}^{(T)} [logL(\theta; X, Z)] and \theta^{t+1}) = argmax_{\theta} Q(\theta \mid \theta^{(t)}).$$

Farthest First

This algorithm is a fast and greedy algorithm. In summary, k points are initially selected as cluster centers. The first center is randomly selected. The second center is greedily selected as the point furthest from the first, while the remaining centers are obtained by greedily selecting the point farthest from the set of already chosen centers. The remaining data points are added to the cluster whose center is the closest to it [115].

For each $Xi = [x_i, 1, x_i, 2, ..., x_{i,m}]$ in D that is described by m categorical attributes, we use $f(X_{i,j} | D)$ to denote the frequency count of attribute value $x_{i,j}$ in the dataset. A scoring function is designed for evaluating each point, and this is defined as:

$$Score(X_i) = \sum_{j=1}^m f(X_{i,j} \mid D).$$

4.4.2. Fuzzy C-Means

Unlike K-means and other classical clustering algorithms in which data just belongs to a certain cluster, in fuzzy C-means clustering algorithm, data may belong to two or more clusters at certain degree [116]. The degree of membership in fuzzy C-means algorithm can accept any value from the interval [0, 1]. This type of representation is more expressive of the data which includes a better detailed view of the data and relationships among clusters [117]. It may, as well, alleviate the problem associated with K-means clustering, which is basically the assignment of an appropriate cluster to data when it is on the same

distance with respect to center of clusters [118], in which case, the degree of membership assigned by fuzzy C-means is equal for the two clusters.

The fuzzy C-means algorithm as proposed in 1983 is to find fuzzy membership functions which minimize the following cost function [119].

$$J_m(U,R) = \sum_{i=1}^N \sum_{j=1}^K \mu_{ij}^m |x_i - r_j|^2,$$
(1)

subject to:

$$\mu_{ij} \in [0, 1]; \sum_{j=1}^{k} \mu_{ij} = 1 \, \forall i; 0 < \sum_{i=1}^{n} \mu_{ij} < N, \forall N,$$
 (2)

where $X = \{x_i\}_{i=1}^N$ is the set of data points, $U = \{\mu_{ij}\}_{i,j=1}^{NK}$ is the matrix of membership degrees, $k \in N$ is the number of clusters, $R = \{r_i\}_{i=1}^k$ is the set of representatives, and m is a measure of fuzziness of the cluster. In case $m \to 1$, the fuzzy C-means becomes standard K-means method. Moreover, |.| represents the Frobenius norm of its arguments. An iterative algorithm is summarized in a few steps, as follows (Algorithm 1).

Algorithm 1: Calculate fuzzy C-means	
Require: $X = \{x_i\}_{i=1}^N$ and k	
U^0 is randomly initialized	
while $ U^{k+1} - U^k > e$ do	
Calculate r_j as follows.	
r_j	$=\frac{\sum_{i=1}^{n}\mu_{ij}^{m}x_{i}}{\sum_{i=1}^{n}\mu_{ij}^{m}}, \ j=1,,k$
Update u_{ii} as follows.	
2, , , , , , , , , , , , , , , , , , ,	$r_{ij} = rac{1}{\sum_{i=1}^C \left(\left rac{x_i - r_j}{x_i - r_k} ight ight)^{rac{2}{m-1}}}$
end while	

End

Fuzzy partition coefficient (FPC) is used to find the optimal number of the clusters associated with fuzzy c-means algorithm. This algorithm measures the amount of overlap between clusters. The resulting coefficient belongs to the interval [0, 1] with its maximum value at *one* corresponding to no overlap between clusters. Fuzzy partition coefficient method is a simple yet effective algorithm which computes the coefficients associated with fuzzy clustering with certain number of clusters as follows [119].

$$FPC = \frac{1}{N} \sum_{i=1}^{N} \sum_{j=1}^{K} \mu_{ij}^{2}.$$
(3)

This index is computed for different number of clusters, and the number of clusters corresponding to maximum number of FPC is considered as the optimal value of clusters for performing fuzzy C-Means.

After performing clustering, each cluster is assigned to the class whose members mostly belong to. In this way, multiple clusters represent a single class. Since the distribution of data in each class may not be just around a single mean value, using multiple clusters makes it possible to find multiple separate points around which data corresponding to a class is distributed.

4.4.3. Fuzzy Partition Coefficient

As mentioned earlier, FPC is used for finding the optimal number of clusters for fuzzy clustering. This index is iterated versus different number of clusters. The results of this experiment are depicted in Figure 6. Higher FPC corresponds to better fuzzy C-means prediction. As can be seen from this figure, the optimal number of clusters need to be considered using fuzzy C-means is 10.



Figure 6. Fuzzy partition coefficient versus number of partitions.

4.4.4. Cluster Class Assignment

Similar to the procedure performed under WEKA for classification, the fuzzy C-means algorithm ignores class labels during training, and it builds clusters based on similarity between features, as described earlier. During the testing phase, new data points are assigned to clusters based on the fuzzy C-means learned model from the training phase—with similar data points clustered together. Predicted class labels are then assigned to each cluster, based on the majority value of the class labels associated with the data points assigned to each cluster. The corresponding confusion matrix is then derived from the real labels and the predicted ones, from which the evaluation metrics (precision, recall, and f-measure) described in Section 4.5 are measured.

4.5. Evaluation Metrics

Maheshwari et al. [120] used F-measure and the area under a ROC (Receiver Operating Characteristic) curve (AUC) measures for evaluating classifiers with imbalanced data set because accuracy can be misleading when used to evaluate algorithms on imbalanced data sets.

On the other hand, Shepperd et al. [121] emphasized that "determining classification performance is more subtle than it might first appear since we need to take into account both chance aspects of a classifier (even guessing can lead to some correct classifications) and also what are termed unbalanced data sets where the distribution of classes is far from 50:50".

Since we are dealing with a multi-class (sitting, vibrating, walking, etc.) classification problem, we adapt the evaluation measures used by Maheshwari et al. [120] and the Matthews Correlation Coefficient (*MCC*) for dealing with imbalanced data (unequal distribution of classes) [121], taking the average measures for all classes as we cannot guarantee an even distribution of instances or occurrences per product usage activity or class in the wild.

The performance measures used in this study are described as follows:

• *F*-measure: The *F*-measure performance metric uses recall and precision for performance measurement, and it is a harmonic mean between them. In practice, high *F*-measure value ensures that both recall and precision are reasonably high [120]. Precision is used to measure the exactness of the prediction set, as well as recall its completeness. These can be represented mathematically as:

$$Precision = \frac{TP}{TP + FP'},\tag{4}$$

$$Recall = \frac{TP}{TP + FN'}$$
(5)

$$F-measure = \frac{2*Precision*Recall}{Precision+Recall},$$
(6)

where: TP = True Positives; TN = True Negatives; FP = False Positives; FN = False Negatives.

 Matthews correlation coefficient (MCC): MCC is a correlation coefficient between the observed and predicted classifications. This measure takes into account *TP*, *FP*, *TN*, and *FN* values and is generally regarded as a balanced measure, which can be used even if the classes are unbalanced. This measure returns a value between -1 and +1. A coefficient of +1 represents a perfect prediction, 0 means a random prediction, and -1 indicates total disagreement between prediction and observations [121,122]. The MCC is computed as follows:

$$MCC = \frac{(TP * TN) - (FP * FN)}{\sqrt{(TP + FP)(TP + FN)(TN + FP)(TN + FN)}}.$$
(7)

5. Results

Table 2 presents the empirical results from the unsupervised machine learning experiments. It shows the Precision, Recall, F-measure, and *MCC* for each of the six machine learning techniques used. The applied algorithms generated 6 clusters of similar data instances during the training phase. And in the testing phase, the algorithms attempted a 1-1 mapping of each of the actual six classes/labels shown in Table 3 to each of the generated or predicted (six) clusters. The accuracy metrics were then generated based on the correctness of this mapping.

	Precision	Recall	F-Measure	MCC	Classes Mapped to Clusters
K-means	0.58	0.57	0.57	0.45	All 6
Expectation-maximization (EM)	0.79	0.80	0.79	0.78	All 6
Farthest first	0.57	0.67	0.61	0.31	Only 3
Fuzzy C-means	0.87	0.87	0.87	0.84	All 6

Evaluation Metrics	Class #1 Dropping and Picking Up	Class #2 Sitting	Class #3 Standing	Class #4 Stationary on Side Table	Class #5 Vibrating Surface	Class #6 Walking
Precision	0.94	0.82	0.84	0.76	0.99	0.94
Recall	1.	1.	0.69	0.99	0.53	1.
F-measure	0.97	0.90	0.76	0.86	0.69	0.97
MCC	0.96	0.88	0.72	0.84	0.69	0.96

Table 3. Precision, Recall, F-measure, and MCC for fuzzy C-means.

Based on the last column in Table 2, the comparison results between the classical clustering approaches of K-means, EM and farthest first shows that only the K-means and EM clustering algorithms are able to map all 6 classes to the 6 clusters generated during the training phase. The farthest first algorithm is unable to map all classes to the 6 clusters given the high proportion of false positive and false negatives derived during the training phase of the algorithm as compared to the total number of instances in the dataset.

Furthermore, looking at the F-measure and MCC of the K-means and EM unsupervised machine learning algorithms, we can deduce that the EM algorithm (with F-measure of 79% and MCC of 0.78) outperforms the k-means algorithm (with F-measure of 57% and MCC of 0.45), both in terms of F-measure and MCC. The K-means invariably outperforms the Farthest first algorithm, given that the Farthest first algorithm was only able to correctly cluster or map 3 out of the 6 classes from the dataset to clusters, and resulted in a lower number of correctly classified instances overall (or of all 3 algorithms applied).

The results obtained for fuzzy C-means for clustering are reported with the number of clusters equal to 10 as previously obtained from FPC method in Section 4.4.3. Considering the results associated with fuzzy C-means algorithm in Table 2, it can be concluded that this approach results in superior performance when compared to the classical clustering approaches of K-means, EM and farthest first. Furthermore, the optimal number of clusters estimated in this paper for fuzzy C-means algorithm is 10, which is multiple clusters per class. This means that fuzzy C-means can effectively cluster data samples in cases where data associated with a class are distributed in multiple separate clusters.

For a deeper evaluation of the best performing algorithm, Table 3 presents the break down of the results obtained for fuzzy C-means in terms of evaluation indexes of precision, recall, f-measure, and MCC for each class, separately. It can be observed that the activities of dropping and picking up, sitting, and walking, which correspond to the classes of 1, 2, and 6, are recognized with higher performance than the other three classes of standing, stationary on side table and vibrating on surface, which correspond to classes number 3, 4, and 5. This could be because the clusters generated by fuzzy C-means for movement characteristics observed by sensors for the cases of dropping and picking up, sitting, and walking are more easily separated than the rest of classes, making them easier to be distinguished. In other words, the movement characteristics observed by sensors for two classes of standing and vibrating on a surface have more similarity to each other, making it more difficult for the software to recognize such activities, which may, in turn, also explain why the fuzzy clustering method has improved on other more static approaches.

6. Conclusions

Four different clustering algorithms were applied in this study for the clustering of a sensor-generated product use activity dataset. K-means algorithm is a simple and easy way to classify datasets through assumed k clusters with fixed a priori. EM algorithm is based on the probability or the likelihood of an instance to belong to a cluster following the expectation and maximization steps. The Farthest First algorithm is an iterative, fast and greedy algorithm in which the center of the first cluster is selected at random. Other than these classical clustering techniques, fuzzy C-means algorithm is also used for performing the clustering to test our assumption that results would improve if we enabled multiactivity clusters, where activities belong to multiple clusters to different degrees during real-time behavior/experiences. The highest accuracy (in terms of both the F-measure and MCC model evaluation metrics for dealing with imbalanced datasets) obtained is 0.87 (F-measure) and 84% (MCC) for the fuzzy C-means algorithm—outperforming the other three algorithms applied on the dataset. We also show that the ability to classify different activity types is consistently high. There are no activities that the best performing model struggled to detect.

The results demonstrate that the detection of "unknown" product use activity in the wild from in-built activity sensors using unsupervised machine learning algorithms is a feasible approach. This could be applied practically to achieve the first stage of the "Chatty Factories" concept—product "experience" collection from product use in the wild, and automated methods to detect product usage activities. This allows the product design process to move beyond traditional customer feedback or manual consumer research, to include augmented information about how the product is actually used based on usagedata. This advances the manufacturing sector's ability to redesign products such that they are better fit for purpose. Moreover, product refinement can be accelerated using the proposed methods, as designers will become less dependent on longer-term studies, such as consumer market research, from small samples of consumers. Product usage data gathered using our proposed methods can provide near real-time insights from consumers at scale—across time, space, and social cultures, enabling continuous product refinement.

Author Contributions: Conceptualization, M.L., N.A., M.A.K., P.B. and D.T.B.; methodology, M.L., N.A., M.A.K.; software, M.L., N.A., M.A.K.; validation, M.L., N.A., M.A.K., P.B. and D.T.B.; formal analysis, M.L., N.A., M.A.K., P.B. and D.T.B.; investigation, M.L., N.A., M.A.K., P.B. and D.T.B.; resources, M.L., N.A., M.A.K.; data curation, M.L., N.A., M.A.K.; writing—original draft preparation, M.L., N.A., M.A.K., P.B. and D.T.B.; writing—review and editing, M.L., N.A., M.A.K., P.B. and D.T.B.; visualization, M.L., N.A., M.A.K.; supervision, P.B. and D.T.B.; project administration, P.B. and D.T.B.; funding acquisition, P.B. and D.T.B. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Engineering and Physical Sciences Research Council (EPSRC)—projects EP/R021031/1—Chatty Factories.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Dataset is available at: https://doi.org/10.6084/m9.figshare.11475252 .v1, accessed on 3 December 2020.

Acknowledgments: This research was funded by the Engineering and Physical Sciences Research Council (EPSRC) under the New Industrial Systems theme: Chatty Factories,—EP/R021031/1.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Cordeiro, G.A.; Ordóñez, R.E.C.; Ferro, R. Theoretical proposal of steps for the implementation of the Industry 4.0 concept. *Braz. J. Oper. Prod. Manag.* 2019, *16*, 166–179. [CrossRef]
- Popkova, E.G.; Ragulina, Y.V.; Bogoviz, A.V. Fundamental Differences of Transition to Industry 4.0 from Previous Industrial Revolutions; Springer: Cham, Switzerland, 2019; pp. 21–29. [CrossRef]
- 3. Yin, Y.; Stecke, K.E.; Li, D. The evolution of production systems from Industry 2.0 through Industry 4.0. *Int. J. Prod. Res.* 2018, 56, 848–861. [CrossRef]
- Muhuri, P.K.; Shukla, A.K.; Abraham, A. Industry 4.0: A bibliometric analysis and detailed overview. *Eng. Appl. Artif. Intell.* 2019, 78, 218–235. [CrossRef]
- Lützenberger, J.; Klein, P.; Hribernik, K.; Thoben, K.D. Improving Product-Service Systems by Exploiting Information from the Usage Phase. A Case Study. *Procedia CIRP* 2016, 47, 376–381. [CrossRef]
- Kong, X.T.; Luo, H.; Huang, G.Q.; Yang, X. Industrial wearable system: The human-centric empowering technology in Industry 4.0. J. Intell. Manuf. 2019, 30, 2853–2869. [CrossRef]
- 7. Thürer, M.; Pan, Y.H.; Qu, T.; Luo, H.; Li, C.D.; Huang, G.Q. Internet of Things (IoT) driven kanban system for reverse logistics: solid waste collection. *J. Intell. Manuf.* **2019**, *30*, 2621–2630. [CrossRef]
- 8. Opresnik, D.; Hirsch, M.; Zanetti, C.; Taisch, M. Information—The hidden value of servitization. In *IFIP Advances in Information and Communication Technology*; Springer: New York, NY, USA, 2013; Volume 415, pp. 49–56. [CrossRef]

- 9. Hou, L.; Jiao, R.J. Data-informed inverse design by product usage information: A review, framework and outlook. *J. Intell. Manuf.* **2020**, *31*, 529–552. [CrossRef]
- 10. Yannou, B.; Yvars, P.A.; Hoyle, C.; Chen, W. Set-based design by simulation of usage scenario coverage. *J. Eng. Des.* 2013, 24, 575–603. [CrossRef]
- Zhang, Y.; Li, X.; Zhang, J.; Chen, S.; Zhou, M.; Farneth, R.A.; Marsic, I.; Burd, R.S. CAR—A deep learning structure for concurrent activity recognition. In Proceedings of the 2017 16th ACM/IEEE International Conference on Information Processing in Sensor Networks, IPSN 2017, Pittsburgh, PA, USA, 18–21 April 2017; pp. 299–300. [CrossRef]
- 12. Zheng, P.; Xu, X.; Chen, C.H. A data-driven cyber-physical approach for personalised smart, connected product co-development in a cloud-based environment. *J. Intell. Manuf.* **2020**, *31*, 3–18. [CrossRef]
- Burnap, P.; Branson, D.; Murray-Rust, D.; Preston, J.; Richards, D.; Burnett, D.; Edwards, N.; Firth, R.; Gorkovenko, K.; Khanesar, M.; et al. Chatty factories: A vision for the future of product design and manufacture with IoT. In Proceedings of the Living in the Internet of Things (IoT 2019), London, UK, 1–2 May 2019; pp. 4–6. [CrossRef]
- 14. Crabtree, A.; Chamberlain, A.; Grinter, R.E.; Jones, M.; Rodden, T.; Rogers, Y. Introduction to the Special Issue of "The Turn to The Wild". ACM Trans. Comput. Hum. Interact. 2013, 20. [CrossRef]
- 15. Voet, H.; Altenhof, M.; Ellerich, M.; Schmitt, R.H.; Linke, B. A Framework for the Capture and Analysis of Product Usage Data for Continuous Product Improvement. *J. Manuf. Sci. Eng. Trans. ASME* **2019**, *141*. [CrossRef]
- 16. Moraes, M.M.; Mendes, T.T.; Arantes, R.M.E. Smart Wearables for Cardiac Autonomic Monitoring in Isolated, Confined and Extreme Environments: A Perspective from Field Research in Antarctica. *Sensors* **2021**, *21*, 1303. [CrossRef] [PubMed]
- 17. Milwaukee-Tool. ONE-KEY Tool Tracking, Customization and Security Technology. Available online: https://onekey1.milwaukeetool. com/ (accessed on 24 May 2021).
- 18. Swan, M. Sensor Mania! The Internet of Things, Wearable Computing, Objective Metrics, and the Quantified Self 2.0. *J. Sens. Actuator Netw.* **2012**, *1*, 217–253. [CrossRef]
- Schmitt, R.; IIF—Institut f
 ür Industriekommunikation und Fachmedien GmbH; Apprimus Verlag; Business Forum Qualit
 ät
 (20.:2016:Aachen) Herausgebendes Organ. Smart Quality—QM im Zeitalter von Industrie 4.0: 20. Business Forum Qualit
 ät; 12. und
 13 September 2016, Aachen; Apprimus Verlag: Aachen, Germany, 2016.
- Wang, J.; Chen, Y.; Hao, S.; Peng, X.; Hu, L. Deep learning for sensor-based activity recognition: A survey. *Pattern Recognit. Lett.* 2019, 119, 3–11. [CrossRef]
- Vepakomma, P.; De, D.; Das, S.K.; Bhansali, S. A-Wristocracy: Deep learning on wrist-worn sensing for recognition of user complex activities. In Proceedings of the 2015 IEEE 12th International Conference on Wearable and Implantable Body Sensor Networks, BSN 2015, Cambridge, MA, USA, 9–12 June 2015. [CrossRef]
- 22. Kim, Y.; Toomajian, B. Hand Gesture Recognition Using Micro-Doppler Signatures with Convolutional Neural Network. *IEEE Access* 2016, *4*, 7125–7130. [CrossRef]
- 23. Hammerla, N.Y.; Halloran, S.; Ploetz, T. Deep, Convolutional, and Recurrent Models for Human Activity Recognition Using Wearables. 2016. Available online: http://xxx.lanl.gov/abs/1604.08880 (accessed on 3 December 2020).
- 24. Qin, J.; Liu, L.; Zhang, Z.; Wang, Y.; Shao, L. Compressive Sequential Learning for Action Similarity Labeling. *IEEE Trans. Image Process.* 2016, 25, 756–769. [CrossRef]
- 25. Patel, A.A. Hands-On Unsupervised Learning Using Python: How to Build Applied Machine Learning Solutions from Unlabeled Data; O'Reilly Media: Sebastopol, CA, USA, 2019.
- 26. Ankita.; Rani, S.; Babbar, H.; Coleman, S.; Singh, A.; Aljahdali, H.M. An Efficient and Lightweight Deep Learning Model for Human Activity Recognition Using Smartphones. *Sensors* **2021**, *21*, 3845. [CrossRef]
- 27. Russell, B.; McDaid, A.; Toscano, W.; Hume, P. Moving the Lab into the Mountains: A Pilot Study of Human Activity Recognition in Unstructured Environments. *Sensors* **2021**, *21*, 654. [CrossRef]
- Jun, K.; Choi, S. Unsupervised End-to-End Deep Model for Newborn and Infant Activity Recognition. Sensors 2020, 20, 6467. [CrossRef] [PubMed]
- 29. Gao, X.; Luo, H.; Wang, Q.; Zhao, F.; Ye, L.; Zhang, Y. A Human Activity Recognition Algorithm Based on Stacking Denoising Autoencoder and LightGBM. *Sensors* **2019**, *19*, 947. [CrossRef] [PubMed]
- Wyatt, D.; Philipose, M.; Choudhury, T. Unsupervised Activity Recognition Using Automatically Mined Common Sense. AAAI 2005, 1, 21–27.
- 31. Huynh, D.T.G. Human Activity Recognition with Wearable Sensors. Ph.D. Thesis, Technische Universität, Darmstadt, Germany, 2008.
- 32. Li, F.; Dustdar, S. Incorporating Unsupervised Learning in Activity Recognition. In Proceedings of the AAAI Publications, Workshops at the Twenty-Fifth AAAI Conference on Artificial Intelligence, San Francisco, CA, USA, 7–8 August 2011; pp. 38–41.
- Vandewiele, F.; Motamed, C. An unsupervised learning method for human activity recognition based on a temporal qualitative model. In Proceedings of the International Workshop on Behaviour Analysis and Video Understanding, Antipolis, France, 23 September 2011; p. 9.
- 34. Trabelsi, D.; Mohammed, S.; Chamroukhi, F.; Oukhellou, L.; Amirat, Y. An unsupervised approach for automatic activity recognition based on Hidden Markov Model regression. *IEEE Trans. Autom. Sci. Eng.* **2013**, *10*, 829–835. [CrossRef]
- Uddin, M.Z.; Thang, N.D.; Kim, J.T.; Kim, T. Human Activity Recognition Using Body Joint-Angle Features and Hidden Markov Model. *ETRI J.* 2011, 33, 569–579. [CrossRef]

- 36. Mathie, M.J.; Coster, A.C.F.; Lovell, N.H.; Celler, B.G. Accelerometry: Providing an integrated, practical method for long-term, ambulatory monitoring of human movement. *Physiol. Meas.* **2004**, *25*, R1–R20. [CrossRef] [PubMed]
- 37. Veltink, P.; Bussmann, H.; de Vries, W.; Martens, W.L.J.; Van Lummel, R. Detection of static and dynamic activities using uniaxial accelerometers. *IEEE Trans. Rehabil. Eng.* **1996**, *4*, 375–385. [CrossRef] [PubMed]
- Aminian, K.; Robert, P.; Buchser, E.E.; Rutschmann, B.; Hayoz, D.; Depairon, M. Physical activity monitoring based on accelerometry: Validation and comparison with video observation. *Med. Biol. Eng. Comput.* 1999, 37, 304–308. [CrossRef] [PubMed]
- 39. Foerster, F.; Fahrenberg, J. Motion pattern and posture: Correctly assessed by calibrated accelerometers. *Behav. Res. Methods Instrum. Comput.* **2000**, *32*, 450–457. [CrossRef]
- 40. Isaia, C.; McNally, D.S.; McMaster, S.A.; Branson, D.T. Effect of mechanical preconditioning on the electrical properties of knitted conductive textiles during cyclic loading. *Text. Res. J.* 2019, *89*, 445–460. [CrossRef]
- 41. Cappozzo, A.; Della Croce, U.; Leardini, A.; Chiari, L. Human movement analysis using stereophotogrammetry: Part 1: Theoretical background. *Gait Posture* **2005**, *21*, 186–196. [PubMed]
- 42. Sutherland, D.H. The evolution of clinical gait analysis part I: Kinesiological EMG. Gait Posture 2001, 14, 61–70. [CrossRef]
- 43. Yang, C.C.; Hsu, Y.L.; Shih, K.S.; Lu, J.M. Real-time gait cycle parameter recognition using a wearable accelerometry system. *Sensors* **2011**, *11*, 7314–7326. [CrossRef] [PubMed]
- 44. Zheng, H.; Black, N.D.; Harris, N.D. Position-sensing technologies for movement analysis in stroke rehabilitation. *Med. Biol. Eng. Comput.* 2005, 43, 413–420. [CrossRef] [PubMed]
- 45. Chen, X. Modelling and Predicting Textile Behaviour; Elsevier: Amsterdam, The Netherlands, 2009.
- 46. Almaslukh, B.; Almaslukh, B.; Almuhtadi, J.; Artoli, A. An Effective Deep Autoencoder Approach for Online Smartphone-Based Human Activity Recognition Online Smartphone-Based Human Activity Recognition View Project an Effective Deep Autoencoder Approach for Online Smartphone-Based Human Activity Recognition. Available online: https://www.semanticscholar.org/ paper/An-effective-deep-autoencoder-approach-for-online-Almaslukh-Artoli/d77c62ee69df0bdbd0be2a95001e1a03f6228cc1 (accessed on 24 May 2021).
- Alsheikh, M.A.; Selim, A.; Niyato, D.; Doyle, L.; Lin, S.; Tan, H.P. Deep Activity Recognition Models with Triaxial Accelerometers. In Proceedings of the Workshops at the Thirtieth AAAI Conference on Artificial Intelligence, Phoenix, AZ, USA, 12–17 February 2016; pp. 8–13.
- 48. LSTM Networks for Mobile Human Activity Recognition; Atlantis Press: Paris, France, 2016. [CrossRef]
- 49. Guan, Y.; Ploetz, T. Ensembles of Deep LSTM Learners for Activity Recognition Using Wearables. 2017. Available online: http://xxx.lanl.gov/abs/1703.09370 (accessed on 5 January 2021). [CrossRef]
- 50. Fang, H.; Hu, C. Recognizing human activity in smart home using deep learning algorithm. In Proceedings of the 33rd Chinese Control Conference, CCC 2014, Nanjing, China, 28–30 July 2014; pp. 4716–4720. [CrossRef]
- Murad, A.; Pyun, J.Y. Deep Recurrent Neural Networks for Human Activity Recognition. Sensors 2017, 17, 2556. [CrossRef] [PubMed]
- Gjoreski, H.; Bizjak, J.; Gjoreski, M.; Gams, M. Comparing Deep and Classical Machine Learning Methods for Human Activity Recognition using Wrist Accelerometer. In Proceedings of the IJCAI-16 Workshop on Deep Learning for Artificial Intelligence (DLAI), New York, NY, USA, 9–15 July 2016; pp. 1–7.
- Ha, S.; Choi, S. Convolutional neural networks for human activity recognition using multiple accelerometer and gyroscope sensors. In Proceedings of the International Joint Conference on Neural Networks, Vancouver, BC, Canada, 24–29 July 2016; pp. 381–388. [CrossRef]
- 54. Inoue, M.; Inoue, S.; Nishida, T. Deep recurrent neural network for mobile human activity recognition with high throughput. *Artif. Life Robot.* **2018**, *23*, 173–185. [CrossRef]
- Jiang, W.; Yin, Z. Human activity recognition using wearable sensors by deep convolutional neural networks. In Proceedings of the MM 2015—ACM Multimedia Conference, Brisbane, Australia, 26–30 October 2015; pp. 1307–1310. [CrossRef]
- Lane, N.D.; Georgiev, P. Can deep learning revolutionize mobile sensing? In Proceedings of the HotMobile 2015—16th International Workshop on Mobile Computing Systems and Applications, Santa Fe, NM, USA, 12–13 February 2015; pp. 117–122. [CrossRef]
- 57. Ordóñez, F.; Roggen, D. Deep Convolutional and LSTM Recurrent Neural Networks for Multimodal Wearable Activity Recognition. *Sensors* **2016**, *16*, 115. [CrossRef] [PubMed]
- Radu, V.; Lane, N.D.; Bhattacharya, S.; Mascolo, C.; Marina, M.K.; Kawsar, F. Towards multimodal deep learning for activity recognition on mobile devices. In Proceedings of the UbiComp 2016 Adjunct—The 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing, Heidelberg, Germany, 12–16 September 2016; pp. 185–188. [CrossRef]
- Ravi, D.; Wong, C.; Lo, B.; Yang, G.Z. Deep learning for human activity recognition: A resource efficient implementation on low-power devices. In Proceedings of the BSN 2016—13th Annual Body Sensor Networks Conference, San Francisco, CA, USA, 14–17 June 2016; pp. 71–76. [CrossRef]
- 60. Ronao, C.A.; Cho, S.B. Deep convolutional neural networks for human activity recognition with smartphone sensors. *Proc. Küse Korea Comput. Congr.* **2015**, *9492*, 858–860. [CrossRef]
- 61. Ronao, C.A.; Cho, S.B. Human activity recognition with smartphone sensors using deep learning neural networks. *Expert Syst. Appl.* **2016**, *59*, 235–244. [CrossRef]

- 62. Walse, K.H.; Dharaskar, R.V.; Thakare, V.M. PCA based optimal ANN classifiers for human activity recognition using mobile sensors data. In *Smart Innovation, Systems and Technologies;* Springer: Berlin/Heidelberg, Germany, 2016; Volume 50, pp. 429–436. [CrossRef]
- Yao, S.; Hu, S.; Zhao, Y.; Zhang, A.; Abdelzaher, T. DeepSense: A unified deep learning framework for time-series mobile sensing data processing. In Proceedings of the 26th International World Wide Web Conference, WWW 2017, Perth, Australia, 3–7 April 2017; International World Wide Web Conferences Steering Committee: Geneva, Switzerland, 2017; pp. 351–360. [CrossRef]
- 64. Zebin, T.; Scully, P.J.; Ozanyan, K.B. Human activity recognition with inertial sensors using a deep learning approach. In Proceedings of the IEEE Sensors, Orlando, FL, USA, 30 October–3 November 2016. [CrossRef]
- Zeng, M.; Nguyen, L.T.; Yu, B.; Mengshoel, O.J.; Zhu, J.; Wu, P.; Zhang, J. Convolutional Neural Networks for human activity recognition using mobile sensors. In Proceedings of the 2014 6th International Conference on Mobile Computing, Applications and Services, MobiCASE 2014, Austin, TX, USA, 6–7 November 2014; pp. 197–205. [CrossRef]
- 66. Zhang, L.; Wu, X.; Luo, D. Real-time activity recognition on smartphones using deep neural networks. In Proceedings of the 2015 IEEE 12th Intl Conf on Ubiquitous Intelligence and Computing and 2015 IEEE 12th Intl Conf on Autonomic and Trusted Computing and 2015 IEEE 15th Intl Conf on Scalable Computing and Communications and Its Associated Workshops (UIC-ATC-ScalCom), Beijing, China, 10–14 August 2015; pp. 1236–1242. [CrossRef]
- Zhang, L.; Wu, X.; Luo, D. Recognizing human activities from raw accelerometer data using deep neural networks. In Proceedings of the 2015 IEEE 14th International Conference on Machine Learning and Applications, ICMLA 2015, Miami, FL, USA, 9–11 December 2015; pp. 865–870. [CrossRef]
- Krishnan, N.C.; Panchanathan, S. Analysis of low resolution accelerometer data for continuous human activity recognition. In Proceedings of the 2008 IEEE International Conference on Acoustics, Speech and Signal Processing, Las Vegas, NV, USA, 31 March–4 April 2008; pp. 3337–3340. [CrossRef]
- 69. Foerster, F.; Smeja, M.; Fahrenberg, J. Detection of posture and motion by accelerometry: A validation study in ambulatory monitoring. *Comput. Hum. Behav.* **1999**, *15*, 571–583. [CrossRef]
- 70. Najafi, B.; Aminian, K.; Paraschiv-Ionescu, A.; Loew, F.; Büla, C.J.; Robert, P. Ambulatory system for human motion analysis using a kinematic sensor: Monitoring of daily physical activity in the elderly. *IEEE Trans. Biomed. Eng.* 2003, 50, 711–723. [CrossRef]
- Banos, O.; Damas, M.; Pomares, H.; Prieto, A.; Rojas, I. Daily living activity recognition based on statistical feature quality group selection. *Expert Syst. Appl.* 2012, 39, 8013–8021. [CrossRef]
- 72. Cheng, W.Y.; Scotland, A.; Lipsmeier, F.; Kilchenmann, T.; Jin, L.; Schjodt-Eriksen, J.; Wolf, D.; Zhang-Schaerer, Y.P.; Garcia, I.F.; Siebourg-Polster, J.; et al. Human Activity Recognition from Sensor-Based Large-Scale Continuous Monitoring of Parkinson's Disease Patients. In Proceedings of the 2017 IEEE 2nd International Conference on Connected Health: Applications, Systems and Engineering Technologies, CHASE 2017, Philadelphia, PA, USA, 17–19 July 2017; pp. 249–250. [CrossRef]
- 73. Bhattacharya, S.; Lane, N.D. From smart to deep: Robust activity recognition on smartwatches using deep learning. In Proceedings of the 2016 IEEE International Conference on Pervasive Computing and Communication Workshops, PerCom Workshops 2016, Sydney, NSW, Australia, 14–18 March 2016; Institute of Electrical and Electronics Engineers Inc.: Sydney, NSW, Australia, 2016. [CrossRef]
- Lee, S.M.; Yoon, S.M.; Cho, H. Human activity recognition from accelerometer data using Convolutional Neural Network. In Proceedings of the 2017 IEEE International Conference on Big Data and Smart Computing, BigComp 2017, Jeju, Korea, 13–16 February 2017; pp. 131–134. [CrossRef]
- 75. Thomas Plötz, N.Y.H. Feature Learning for Activity Recognition in Ubiquitous Computing. In Proceedings of the Twenty-Second International Joint Conference on Artificial Intelligence, Barcelona, Spain, 16–22 July 2011; pp. 1729–1734. [CrossRef]
- Zhang, L.; Wu, X.; Luo, D. Human activity recognition with HMM-DNN model. In Proceedings of the 2015 IEEE 14th International Conference on Cognitive Informatics and Cognitive Computing, ICCI*CC 2015, Beijing, China, 6–8 July 2015; pp. 192–197. [CrossRef]
- 77. Ravi, D.; Wong, C.; Lo, B.; Yang, G.Z. A Deep Learning Approach to on-Node Sensor Data Analytics for Mobile or Wearable Devices. *IEEE J. Biomed. Health Inform.* 2017, 21, 56–64. [CrossRef]
- 78. Zheng, Y.; Liu, Q.; Chen, E.; Ge, Y.; Zhao, J.L. Exploiting multi-channels deep convolutional neural networks for multivariate time series classification. *Front. Comput. Sci.* **2016**, *10*, 96–112. [CrossRef]
- 79. Chen, Y.; Xue, Y. A Deep Learning Approach to Human Activity Recognition Based on Single Accelerometer. In Proceedings of the 2015 IEEE International Conference on Systems, Man, and Cybernetics, SMC 2015, Hong Kong, China, 9–12 October 2015; Institute of Electrical and Electronics Engineers Inc.: Hong Kong, China, 2016; pp. 1488–1492. [CrossRef]
- 80. Edel, M.; Köppe, E. Binarized-BLSTM-RNN based Human Activity Recognition. In Proceedings of the 2016 International Conference on Indoor Positioning and Indoor Navigation, IPIN 2016, Alcala de Henares, Spain, 4–7 October 2016. [CrossRef]
- Mohammed, S.; Tashev, I. Unsupervised deep representation learning to remove motion artifacts in free-mode body sensor networks. In Proceedings of the 2017 IEEE 14th International Conference on Wearable and Implantable Body Sensor Networks, BSN 2017, Eindhoven, The Netherlands, 9–12 May 2017; pp. 183–188. [CrossRef]
- Panwar, M.; Ram Dyuthi, S.; Chandra Prakash, K.; Biswas, D.; Acharyya, A.; Maharatna, K.; Gautam, A.; Naik, G.R. CNN based approach for activity recognition using a wrist-worn accelerometer. In Proceedings of the Annual International Conference of the IEEE Engineering in Medicine and Biology Society, EMBS, Jeju, Korea, 11–15 July 2017; pp. 2438–2441. [CrossRef]

- Mantyjarvi, J.; Himberg, J.; Seppanen, T. Recognizing human motion with multiple acceleration sensors. In Proceedings of the 2001 IEEE International Conference on Systems, Man and Cybernetics. e-Systems and e-Man for Cybernetics in Cyberspace (Cat.No.01CH37236), Tucson, AZ, USA, 7–10 October 2001; Volume 2, pp. 747–752. [CrossRef]
- 84. Bao, L.; Intille, S.S. Activity Recognition from User-Annotated Acceleration Data; Springer: Berlin/Heidelberg, Germany, 2004; pp. 1–17. [CrossRef]
- Banos, O.; Galvez, J.M.; Damas, M.; Pomares, H.; Rojas, I. Window Size Impact in Human Activity Recognition. Sensors 2014, 14, 6474–6499. [CrossRef] [PubMed]
- 86. Hammerla, N.Y.; Fisher, J.M.; Andras, P.; Rochester, L.; Walker, R.; Plötz, T. PD disease state assessment in naturalistic environments using deep learning. *Proc. Natl. Conf. Artif. Intell.* **2015**, *3*, 1742–1748.
- Hayashi, T.; Nishida, M.; Kitaoka, N.; Takeda, K. Daily activity recognition based on DNN using environmental sound and acceleration signals. In Proceedings of the 2015 23rd European Signal Processing Conference, EUSIPCO 2015, Nice, France, 31 August–4 September 2015; pp. 2306–2310. [CrossRef]
- Lane, N.D.; Georgiev, P.; Qendro, L. DeepEar: Robust smartphone audio sensing in unconstrained acoustic environments using deep learning. In Proceedings of the UbiComp 2015—The 2015 ACM International Joint Conference on Pervasive and Ubiquitous Computing, Osaka, Japan, 7–11 September 2015; pp. 283–294. [CrossRef]
- Sathyanarayana, A.; Joty, S.; Fernandez-Luque, L.; Ofli, F.; Srivastava, J.; Elmagarmid, A.; Taheri, S.; Arora, T. Impact of Physical Activity on Sleep: A Deep Learning Based Exploration. *arXiv* 2016, arXiv:1607.07034.
- Kim, Y.; Li, Y. Human Activity Classification with Transmission and Reflection Coefficients of On-Body Antennas Through Deep Convolutional Neural Networks. *IEEE Trans. Antennas Propag.* 2017, 65, 2764–2768. [CrossRef]
- Liu, C.; Zhang, L.; Liu, Z.; Liu, K.; Li, X.; Liu, Y. Lasagna: Towards deep hierarchical understanding and searching over mobile sensing data. In Proceedings of the Annual International Conference on Mobile Computing and Networking, MOBICOM, New York, NY, USA, 3–7 October 2016; Association for Computing Machinery: New York, NY, USA, 2016; pp. 334–347. [CrossRef]
- 92. Wang, J.; Zhang, X.; Gao, Q.; Yue, H.; Wang, H. Device-Free Wireless Localization and Activity Recognition: A Deep Learning Approach. *IEEE Trans. Veh. Technol.* 2017, *66*, 6258–6267. [CrossRef]
- Costello, S. *The Sensors That Make the iPhone So Cool*. Available online: https://www.lifewire.com/sensors-that-make-iphone-socool-2000370 (accessed on 4 February 2020).
- 94. Allan, A. Basic Sensors in iOS; O'Reilly Media: Sebastopol, CA, USA, 2011; p. 89.
- MATLAB. 7 Reasons MATLAB Is the Easiest and Most Productive Environment for Engineers and Scientists. 2020. Available online: https://www.linkedin.com/pulse/7-reasons-matlab-easiest-most-productive-environment-engineers-tate?articleId= 6228200317306044416 (accessed on 4 February 2020).
- 96. Frank, E.; Hall, M.; Trigg, L.; Holmes, G.; Witten, I.H. Data mining in bioinformatics using Weka. *Bioinformatics* 2004, 20, 2479–2481. [CrossRef]
- 97. Hall, M.; Frank, E.; Holmes, G.; Pfahringer, B.; Reutemann, P.; Witten, I.H. The WEKA data mining software: An update. ACM SIGKDD Explor. Newsl. 2009, 11, 10–18. [CrossRef]
- Malhotra, R.; Khanna, M. Examining the effectiveness of machine learning algorithms for prediction of change prone classes. In Proceedings of the 2014 International Conference on High Performance Computing & Simulation (HPCS), Bologna, Italy, 21–25 July 2014; pp. 635–642.
- 100. Malhotra, R.; Khanna, M. Investigation of relationship between object-oriented metrics and change proneness. *Int. J. Mach. Learn. Cybern.* **2013**, *4*, 273–286. [CrossRef]
- 101. Nizamani, S.; Memon, N.; Wiil, U.K.; Karampelas, P. Modeling suspicious email detection using enhanced feature selection. *arXiv* 2013, arXiv:1312.1971.
- 102. Kodinariya, T.M.; Makwana, P.R. Review on determining number of Cluster in K-Means Clustering. Int. J. 2013, 1, 90–95.
- 103. Pollard, D. Strong consistency of k-means clustering. Ann. Stat. 1981, 9, 135–140. [CrossRef]
- 104. Chakrabarty, N.; Rana, S.; Chowdhury, S.; Maitra, R. RBM Based Joke Recommendation System and Joke Reader Segmentation. In *International Conference on Pattern Recognition and Machine Intelligence*; Springer: Cham, Switzerland, 2019; pp. 229–239.
- 105. Kuraria, A.; Jharbade, N.; Soni, M. Centroid Selection Process Using WCSS and Elbow Method for K-Mean Clustering Algorithm in Data Mining. *Int. J. Sci. Res. Sci. Eng. Technol.* 2018, *4*, 190–195. [CrossRef]
- Shippey, T.; Bowes, D.; Hall, T. Automatically identifying code features for software defect prediction: Using AST N-grams. *Inf. Softw. Technol.* 2019, 106, 142–160. [CrossRef]
- 107. Zang, H.; Zhang, S.; Hapeshi, K. A review of nature-inspired algorithms. J. Bionic Eng. 2010, 7, S232–S237. [CrossRef]
- 108. Shrivastava, P.; Shukla, A.; Vepakomma, P.; Bhansali, N.; Verma, K. A survey of nature-inspired algorithms for feature selection to identify Parkinson's disease. *Comput. Methods Programs Biomed.* **2017**, *139*, 171–179. [CrossRef]
- MacQueen, J. Some methods for classification and analysis of multivariate observations. In Proceedings of the 5th Berkeley Symposium on Mathematical Statistics and Probability; University of California Press: Berkeley, CA, USA, 1967; pp. 281–297.
- 110. Han, J.; Kamber, M.; Pei, J. Data Mining: Concepts and Techniques; Elsevier: Amsterdam, The Netherlands, 2012. [CrossRef]

- 111. Witten, I.H.; Frank, E.; Hall, M.A.; Pal, C.J. Data Mining: Practical Machine Learning Tools and Techniques; Elsevier Inc.: Amsterdam, The Netherlands, 2016; pp. 1–621. [CrossRef]
- 112. Likas, A.; Vlassis, N.; Verbeek, J.J. The Global k-Means Clustering Algorithm. Pattern Recognit. 2003, 36, 451–461. [CrossRef]
- 113. Dempster, A.P.; Laird, N.M.; Rubin, D.B. Maximum Likelihood from Incomplete Data Via the *EM* Algorithm. *J. R. Stat. Soc. Ser. B* (*Methodol.*) **1977**, *39*, 1–22. [CrossRef]
- 114. McLachlan, G.J.; Krishnan, T. *The EM Algorithm and Extensions*, 2nd ed.; Wiley Blackwell: Hoboken, NJ, USA, 2007; pp. 1–369. [CrossRef]
- 115. Sharmila, S.; Kumar, M. An optimized farthest first clustering algorithm. In Proceedings of the 2013 Nirma University International Conference on Engineering, NUiCONE 2013, Ahmedabad, India, 28–30 November 2013. [CrossRef]
- 116. a Miyamoto, S.; Ichihashi, H.; Honda, K. *Algorithms for Fuzzy Clustering Methods in c-Means Clustering with Applications*; Springer: Berlin/Heidelberg, Germany, 2008.
- 117. Stetco, A.; Zeng, X.J.; Keane, J. Fuzzy C-means++: Fuzzy C-means with effective seeding initialization. *Expert Syst. Appl.* 2015, 42, 7541–7548. [CrossRef]
- 118. Döring, C.; Lesot, M.J.; Kruse, R. Data analysis with fuzzy clustering methods. *Comput. Stat. Data Anal.* 2006, 51, 192–214. [CrossRef]
- 119. Bezdek, J.C. Pattern Recognition with Fuzzy Objective Function Algorithms; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2013.
- 120. Maheshwari, S.; Agrawal, J.; Sharma, S. A new approach for Classification of Highly Imbalancade Datasets using Evolutionary Algorithms. *Int. J. Sci. Eng. Res.* **2011**, *2*, 1–5.
- 121. Shepperd, M.; Bowes, D.; Hall, T. Researcher bias: The use of machine learning in software defect prediction. *IEEE Trans. Softw. Eng.* **2014**, 40, 603–616. [CrossRef]
- 122. Zhou, Y.; Wursch, M.; Giger, E.; Gall, H.; Lu, J. A bayesian network based approach for change coupling prediction. In Proceedings of the 15th Working Conference on Reverse Engineering, WCRE'08, Antwerp, Belgium, 15–18 October 2008; pp. 27–36.